Author's response to reviews

Title: Weather conditions and sudden sensorineural hearing loss

Authors:

Vasilis Danielides (vdanielidis@hotmail.com)
Christina-Sophia Nousia (bnousias@in.gr)
Aristides Bartzokas (abartzok@cc.uoi.gr)
Christos Lolis (me00061@cc.uoi.gr)
Antonios Skevas (otorl@cc.uoi.gr)

Version: 2 Date: 10 May 2002

To the Editors
BMC Ear Nose and Throat Disorders

Ioannina, May 10th, 2002

Dear Editors

We are submitting a corrected version of our research article entitled "Weather conditions and sudden sensorineural hearing loss" for criticism and publication in BMC Ear Nose and Throat Disorders. We would like to inform you that the modifications made are the following:

1) Title page: correction of affiliations
2) Page 3, line 14: (x2) was changed to (I[double dagger]2)
3) Page 6, line 22: [1,3-5] was changed to [2-5]
4) Page 8, line 16: Confidence Interval was changed Confidence Level
5) Page 8, line 24: (Table 5) was changed to (Table 3)

We apologize for the inconvenience

Kind regards
Dr Vasilis G. Danielides, MD, Assistant Professor
Department of Otorhinolaryngology
University of Ioannina
Medical School
PO Box 1186
Ioannina, 45110, Greece
Tel: 0030 6510 97513
Fax: 0030 6510 77430
E-mail: vdanielidis@hotmail.com